CDE December, 1998

## **Centre for Development Economics**

## Structural Breaks in Consumption Patterns: India, 1952 To 1991

Brinda Viswanathan

Working Paper No. 61

#### ABSTRACT

The tests for structural breaks in consumption patterns indicate multiple break points which are not uniform across the population groups and also across commodity groups. Further, the results indicate that the breaks could often be induced by the changes in the data collection methodology of the survey and not due to changes in consumer behaviour alone. Apart from this, there is a shift in the consumption pattern during the mid-1980's in both the rural and the urban sectors. For the lowest expenditure class the shift is away from food items with the rural sector showing a change in the price response and the urban sector showing a change in the total expenditure coefficient. For the middle and the upper expenditure classes the shifts are not only from the food items towards non food items but also from the 'food' group that includes items like cereals, milk and milk products towards the 'other food' group which includes items like vegetables and fruits. Its causes are found to be changes in preferences as well as the income effect.

<sup>\*</sup> Research Fellow at the Centre for Development Economics, Delhi School of Economics. This work is based on the author's Ph.D. thesis titled 'Some Econometric Issues in Consumer Behaviour Analysis'. The author would like to acknowledge the comments received from an anonymous referee, Dr.K.L.Krishna, Dr. M.H. Suryanarayana, Dr. S. Sarkar, and Dr. R. Krishnan on an earlier draft of this paper .

Moreover, the scope, concept and design of surveys, the way questionnaires are structured and information is elicited have changed over time. These factors affect the comparability of estimates across time and reduce their usefulness for study of changes in consumption patterns.

Rest of the paper is organised as follows. Section 2 discusses briefly the findings of the existing studies on the changes in consumption patterns for India. Section 3 describes the organisation of the data set as used in this analysis. Section 4 discusses the issues involved in the choice of break points for the given sample, the econometric methodology used for estimating and testing a model with structural breaks, and the demand system specification. Section 5 discusses the results for the three-commodity dynamic demand model and the final section concludes the paper.

### 2. Changes In Consumption Patterns: the Indian Case

Since the mid-seventies shifts in consumption patterns for the Indian population have been noted, by Meenakshi (1996a, 1996b); Perspective Planning Division (1993); Radhakrishna (1991); Radhakrishna and Ravi (1990, 1992); and Suryanarayana (1995). All these studies report shifts in consumption patterns (in per capita terms) away from food items in general to non food items, from cereals to noncereal food items within the food group, and from coarse cereals to superior cereals like wheat and rice within the cereal group. However, the choice of variables to address the issue of shifts in consumption patterns and the possible underlying causes vary across these studies.

Radhakrishna and Ravi (1990) (RR, henceforth) have analysed this issue based on a (hierarchic) demand system using the dummy variable approach, for the period 1964-'65 (19<sup>th</sup> round of the NSS) to 1986-'87 (42<sup>nd</sup> round of the NSS) with a predetermined break point in the year 1973-74 (28<sup>th</sup> round of the NSS).<sup>1</sup> The results show the dummy coefficients to be significant for all the expenditure groups and for all the commodities, with a decrease in the marginal budget share (MBS) of cereals and an increase in that of the non-food items during the second period. Further, within the food group there has been a shift from the coarse cereals towards rice and wheat for all the population groups considered by them, and these have been attributed to changes in tastes.

Based on a trend analysis of consumption patterns, Radhakrishna (1991) has observed similar shifts, since mid 1970s, uniformly across all expenditure groups including the the poorer groups in rural areas, and larger shifts in states with higher per capita incomes. The cause for the changes in consumption patterns have been noted as the changes in tastes particularly for the cereal items. This has been based on the observation that the quantity of per capita cereal consumption of the poor had not shown any increase in spite of the improvements in their incomes and decline in relative price of cereals. A further analysis into this aspect in Radhakrishna and Ravi (1992) has indicated adverse effects of the shift away from cereals on the nutritional intake of the rural poor. It has been noted that for the period between 1970-71 and 1987-88, the overall per capita total consumer expenditure improved with increases in non cereal and non food consumption and decreases in the total cereal consumption particularly for the rural sector. The factors that contribute towards the positive effects have been noted as growth and relative price changes and the adverse effect due to taste changes.<sup>2</sup>

<sup>&</sup>lt;sup>1</sup> The first stage of the demand model is at the seven-commodity classification ('cereals', 'milk and milk products', 'edible oil', 'meat, egg and fish' 'sugar, gur, etc.', 'other food' and 'non food') based on the linear expenditure system (LES). Dummies are also introduced at the subgroup level where the 'cereal' group is further classified into 'rice'; 'wheat' and 'other cereals' and the demand model is based on the Nasse's linear expenditure system (NLES). The analysis is carried out for various expenditure groups formed on the basis of the poverty line (PL) adopted by the Government of India. The groups are: below 75% of PL = v. poor; 75% of PL to PL = moderately poor; PL to 150% of PL = mid strata and above 150% of PL = rich.

<sup>&</sup>lt;sup>2</sup> The changes in aggregate per capita consumption and the expenditure on various commodity groups have been decomposed on the basis that the change between two periods may be due to changes in: real mean expenditure (growth component) or inequality (redistributive component) or relative prices or consumer preferences (taste change). The study first estimates a hierarchic demand model as in RR (1990) and has noticed similar results. This study also addresses the effect of such changes on the nutrient intake of the different population groups in both the rural and the urban sectors.

The Expert Group on the Estimation of Proportion and Number of Poor (Perspective Planning Division, 1993) has noted that the shifts in consumption patterns involving reduction in shares of cereals and foodgrains is the cause of reduction in average calorie intake across expenditure classes since the mid seventies. But the Expert Group estimates have been based on consumer expenditure proportions at current prices and this may perhaps exaggerate the changes, as the associated relative price changes have not been accounted for. Suryanarayana (1995) has partly overcome this problem by examining the changes in aggregate consumer expenditure at constant prices with the possible appropriate fractile-group specific price adjustments and commodity specific price adjustments, and changes in size and composition of cereal consumption in physical terms. In this study changes in consumption patterns have been observed since 1977-'78. The change particularly among the poor has been towards a larger variety of food items with a marginal shift towards non cereal items, and towards a better quality with a shift away from coarse cereals towards superior cereals like rice and wheat, within the cereal group. Interestingly enough these changes have been noticed during a period with limited increases in aggregate real consumption and also when the consumption levels happened to be still below the subsistence levels in terms of calorie intake. However, unlike Radhakrishna (1991), these changes in the composition of consumption basket have been attributed partly in response to changing tastes, changing relative prices and their substitution effects on consumer choices, increasing market dependence on superior but costly cereals and decreasing per capita availability of coarse cereals due to a decline in production.<sup>3</sup>

All the studies mentioned above are at the all India level. Meenakshi (1996a) has looked at the trend changes in consumption pattern for five different regions of India (formed by grouping different states) between the period 1972-73 and 1987-88. Broadly the changes are similar to those at the national level but certain differences across regions and quartile groups have been highlighted. The important findings include: operation of Engel's law uniformly across regions and across quartile groups

<sup>&</sup>lt;sup>3</sup> These changes were perhaps due to the changes in the rural labour market which involved a decrease in self-employment and increase in wage employment and growing casualisation of wage labour coupled with the fact that coarse cereals were grown largely for self-consumption whose production had declined.

in each region, decrease in cereal consumption within the food group with the exception of the lowest quartile in the Eastern and the Southern regions of rural India, substitution of costlier cereals like rice and wheat in place of the coarse cereals. The causes of these changes have been primarily attributed to the effect of income changes and in some regions to changes in relative prices. In Meenakshi (1996b), the changes in food consumption has been analysed for different states, based on the LES with time trends either for the 'subsistence' coefficient or the MBS. With a few exceptions, in both the rural and the urban sectors, the expenditure elasticity for the cereals has been found to be lower in the dynamic model.

These studies are important in that they indicate the direction of change for various commodities and possible underlying causes of these shifts. However, the analyses in all these studies are based either on the trends in consumption patterns for various items or the dynamic demand model based on the LES. The former approach is restrictive as it is difficult to dissociate the changes in demand occurring due to changes in income or relative prices (or both), from those occurring due to changes in tastes or other factors. Therefore, a system of demand equations (derived from the neo-classical theory of consumer behaviour) is appropriate to address the problem of changing consumption patterns. Even the existing studies based on demand system give little attention to account for dynamic structure in consumption patterns that are likely to be affected by the various factors mentioned above. Moreover, the functional form chosen to estimate the demand systems are all based on the LES, which is restrictive due to its additively separable preference structure and linear Engel curves.<sup>4</sup>

Keeping in view the gaps in the existing studies this paper tries to improve upon the methodology and the framework to analyse the consumer behaviour in India over a period of time. The objectives of this study are: (i) to test for structural breaks in consumption pattern in a demand system framework, as revealed by the NSS data;

<sup>&</sup>lt;sup>4</sup>As the LES is derived from an additively separable utility function, the price effects are directly proportional to the income effects. This imposes a structure on the price elasticities, independent of the nature of the data set. However, the sub-group model (NLES) in RR performs better in capturing the price responses as the coefficients representing committed quantities in the latter are specified as functions of prices. This captures the substitution effects between commodities appropriately (see., Brown and Deaton, 1972, pp.1197). Further, Majumder (1986), Ray (1986) and Viswanathan (1998) show that the linearity of Engel curves and the additive separability assumption of the LES are restrictive in the Indian context.

(ii) to explore the underlying causes of the changes; and (iii) to re-estimate the demand system incorporating these changes. The issue of shifts in consumption pattern is being addressed at the all-India level for four ordinal groups of population and all-groups combined, at the three-commodity classification; for the rural and the urban sectors separately.

(

1

### 3. Data

The consumer expenditure data from 4<sup>th</sup> (Apr.-Sep. 1952) to 47<sup>th</sup> round (Jul.-Dec. 1991) is used and the price variables are the Wholesale Price Indices.<sup>5</sup> The expenditure data is obtained form various NSS reports and the price series are obtained from Chandolk (1978) and Government of India (1987).

#### 3.1. Commodity groups

The commodity classification in the published NSS reports varies over the rounds. Hence, 'consistent' commodity groups are formed by aggregating substitutes and near substitutes. The choice of the commodity classification is governed by the availability of the data uniformly across all the rounds for rural and urban all-India data. In this paper the analysis is carried out at the *three-commodity* classification comprising:<sup>6</sup>

(1) Food: consisting of (i) cereals and cereal substitutes and gram, (ii) milk and milk products and (iii) meat, egg and fish;

<sup>&</sup>lt;sup>5</sup> Ideally, in estimating the demand system, one should use the prices that the NSS uses to value the expenditure. However, this is not reported and there is no other source which reports data on retail prices for the period of analysis. Another series on prices is the consumer price index but this series is available only from 1964-'65. A series for some of these commodities was constructed by Jain and Minhas, (1991) and Tendulkar and Jain (1993) which is also available only for the later rounds of the NSS. Hence, the Wholesale Price Index is used for the price series.

<sup>&</sup>lt;sup>6</sup> Though the maximum possible commodity disaggregation that could be obtained from published NSS reports is the nine-commodity level, the analysis here is carried out in terms of a three-commodity group specification. This choice is governed largely by data availability considerations since for a system like the AIDS, given the limited sample size the degrees of freedom will be substantially larger for a three-commodity than for a nine-commodity classification. This is particularly important for measuring the price responses and addressing dynamic issues like changes in consumption patterns. However, the tests for structural breaks have also been carried out at the nine-commodity level but could not be analysed in such detail due to the lack of sufficient degrees of freedom as required for such an analysis.

- (2) Other food: consisting of (i) edible oils, (ii) sugar, 'gur', etc. and (iii) miscellaneous other food like fruits and vegetables, spices and condiments, pulses and its products, beverages and refreshments etc. and
- (3) Non food: consisting of (i) clothing and footwear, (ii) fuel and light and (iii) miscellaneous goods and services, inclusive of durable goods.

#### 3.2. Formation of ordinal groups

The consumption data as published by the NSS is in the form of size distribution of households and population across 11 to 14 monthly per capita total consumer expenditure (MPCTCE) classes with corresponding MPCTCE and its break up into a number of broad commodity groups at current prices. These expenditure classes are not updated systematically in keeping with changing prices.<sup>7</sup> Therefore, the average expenditure on a particular item in a given expenditure class will not be comparable over time due to (a) inflation; and (b) varying population frequencies (Suryanarayana, 1991). In order to facilitate comparison of the NSS consumer expenditure distributions, the data set has been reconstructed to form ordinal (population) groups, separately for the rural and the urban sectors. The ordinal groups are defined with reference to total per capita consumer expenditure and are identified by ranking the persons by level of per capita expenditure (PCE); based on quadratic interpolation.

#### The ordinal groups are:

group 1: bottom 30 % (< 30%); group 2: next 20% (30-50 %); group 3: middle 30% (50-80%); group 4: richest 20% (> 80%) top most 20 per cent; and all-groups: all the ordinal classes combined.

The economic reasoning underlying the choice of the ordinal groups is as

<sup>&</sup>lt;sup>7</sup> The NSS expenditure class intervals were kept invariant till the 28<sup>th</sup> rounds (1973-'74) even though the consumer expenditures were collected and published at current prices. The limited changes in class intervals that have been attempted since then are *ad hoc* and are not decided statistically taking into account the accompanying changes in prices. The lack of an 'appropriate' price series for deflation leads to the incomparability of the nominal consumption expenditure classes over time.

follows, Estimates of poverty for India vary between 30 and 50 per cent depending upon the method of estimation (Ahluwalia, 1978, Suryanarayana and Geetha, 1996 and Tendulkar, et al., 1993). However, almost all the studies show that about 30 per cent of the population have been perpetually in poverty. This finding motivated the choice of the bottom 30 per cent as one ordinal group. Further, the studies cited above also show fluctuations in poverty in the range 30 to 50 per cent implying possibly that this group bears the brunt of fluctuations in economic performance. This factor motivated the choice of the second ordinal group as 30 to 50 per cent. These two ordinal groups thus, constitute the 'poorest' and the 'poor' segments respectively of the population for whom a major constraint binding consumer choice and hence consumption pattern would be income. The top 50 per cent of the population is broken up into two groups-middle 30 per cent and the topmost 20 per cent. The former class could be taken to represent the consumption pattern of the 'middle class'; and the latter one with the largest proportion of expenditure spent on non-food items (and hence a larger variety in consumption) would have a distinct pattern of consumption in comparison to all the other groups. Thus, one expects that the formation of population groups in this manner would allow for the consumption pattern to be heterogeneous between groups and homogeneous within each group. The ordinal groups are formed based on the ranking by level of per capita total consumer expenditure based on quadratic interpolation.

(Gh)

data

NS!

at d

que

inst

exț

em

col

da

tin

N!

he

sł

g١

sl

sl

с

1

t

. t 1

## 4. Methodology

## 4.1. Nature of the data set and the choice of regimes

Instead of testing each of the sample point for a break *a priori* information based on the nature of the data set is used to demarcate the different regimes. The literature on the limitations of the database discuss as to how the reported values of consumption for the different items are affected by the changes in the methodology of data collection which in turn would (a) bias the estimates of poverty measures (Suryanarayana, 1996) (b) bias aggregate consumption or consumer expenditure for different commodities (see Minhas, 1988; Mukherjee, 1986; Vaidyanathan, 1986 and the references therein), and (c) bias the Engel elasticities of various commodities (Ghose and Bhattacharya, 1994). These studies further indicate that the use of the NSS data to assess changes over time may be beset with problems. The surveys by the NSSO are not conducted with a view to form a time-series but to construct aggregates at different points of time. Further, the scope, concept and design of surveys, the way questionnaires are structured and information is elicited have changed over time. For instance, changes in the reference period for consumption items, shift from consumer expenditure to multi-purpose integrated household surveys and back, combining employment and consumption enquiries, changes in the schedules, frequency of data collection, seem to affect comparability. Therefore some of these aspects related to the data base would also affect the estimates of a demand system. This study for the first time, tries to examine the effect of changes in methodology of data collection by the NSS from a demand system perspective.

ng

96

er

he

ve

at

Ĵ۶

0

эf

e

n

In this study the first break point is taken at the beginning of the integrated household surveys (IHS, henceforth). This is done on the basis of the plots of budget shares for various commodities, over the NSS rounds as in Viswanathan (1998).<sup>8</sup> For groups 3, 4 and all-groups in both the rural and the urban sector, there is an upward shift in the budget shares of the 'food' group and a downward shift in the budget shares of the 'non food' group for the three-commodity classification. For the nine-commodity classification during the same period, the 'foodgrains' group and the 'miscellaneous non food' group show the upward and downward shifts respectively. In addition, the consumption pattern of group 4 of urban all-India is also marked by an upward shift in the 'other food' shares at the three-commodity level which is due to the shift in shares of the 'miscellaneous other food' group. Therefore, the IHS rounds have been chosen as a separate regime (called as the IHS regime) to be tested for parameter equality with the previous regime.

After these rounds no such unusual shift in the budget shares of some of the commodity groups (for some of the ordinal groups) is observed. However, the data following the IHS are not continuous in time due to the introduction of the quinquennial surveys (QS, henceforth). The rounds 27<sup>th</sup> and 28<sup>th</sup> are chosen as one

<sup>&</sup>lt;sup>8</sup> Mukherjee (1986) observes that during the IHS rounds the aggregate household consumer expenditure is consistently less than the private final consumption estimated in the National Account Statistics.

regime and each of the rounds 32 and 38 is chosen as a separate regime - referred as QS1, QS2 and QS3, respectively. This choice is done to test the possibility of changes in the consumption pattern firstly, due to the data gaps between these surveys which put together cover a period of 15 years and secondly, as they are based on very large samples compared to the other surveys (Ray and Bhattacharya, 1992).

grc

Th

4.4

lir

lir

ch

L

u

T

И

W

ห

р Л

L

1

Finally, with the availability of yearly data after the third quinquennial survey, the rounds 42<sup>nd</sup> to 47<sup>th</sup> are chosen as a separate regime. This regime is referred to as the post-QS3. The schedules for these rounds are similar to the first regime considered above. Moreover, it is to be noted that this regime covers the decade of 1980's when the poverty levels in India were the lowest (Ravallion and Datt, 1996), the annual growth rate of real per capita GDP doubled (Tendulkar and Jain, 1995) and the household savings rate showed large improvements. Simultaneously, during this decade substantial improvements in the per capita consumer expenditures for the poorest deciles with a more diversified consumption basket for all expenditure groups is noted, for both the rural and the urban sectors (Suryanarayana, 1995). It is likely that these economy-wide changes would alter the consumption patterns and hence affect the parameters in the demand model.

Based on the above observations about the database, the anticipated break points are:

Regimes	Rounds	Year
1.Pre-IHS regime	4 <sup>th</sup> - 17 <sup>th</sup>	Apr Sep. 1952 to Sep. 1961 - Jul. 1962
2. IHS regime	$19^{th} - 25^{th}$	Jul, 1967 - Jun, 1968 to Jul, 1970 - Jun, 197
3. Q\$1	27 <sup>th</sup> -28 <sup>th</sup>	Oct. 1972-Sept.'73 to Oct.'73-Jun.'74
4. QS2	32 <sup>nd</sup>	Jun. 1977 - Jun. 1978
5. QS3	38 <sup>th</sup>	Jan Dec. 1983
6. post-QS3	42 <sup>nd</sup> - 47 <sup>th</sup>	Jul. 1986 - Jun. 1987 to Jul Dec. 1991

Table 1. Choice of Regimes Over the NSS Rounds

Note: IHS: Integrated Household Surveys; QS: Quinquennial Surveys

It is important to note that this choice of regimes is *ad hoc* that is, it may or may not determine the final combination of the break points for a particular ordinal group. In order to test the validity of these break points, statistical tests are carried out. The estimation procedure and the tests are described below.

## 4.2. Empirical specification of the demand model

In this study, the functional form used to estimate the demand system is the linear approximate almost ideal demand system (LA-AIDS). The functional form is a linear approximation of the AIDS model of Deaton and Muellbauer (1980).<sup>9</sup> The choice of this specification for the demand system overcomes the limitations of the LES (used in RR) in that it does not have constant marginal budget shares and the underlying preference structure is not limited by additive separability.

The model with dummy variables attached to all the coefficients is as follows:

$$w_{i} = \alpha_{i}^{1} + d\alpha_{i}^{h} D^{h} + \sum_{i=1}^{n} \gamma_{ij}^{1} \log p_{j} + \sum_{j=1}^{n} d\gamma_{ij}^{h} \log F_{j}^{h} + \beta_{i}^{1} \log\left(\frac{Y}{P^{*}}\right) + d\beta_{i}^{h} \log G^{h}$$
(1)

where,  $h \neq l$ 

d as

iges

nich

irge

∕ey,

) as

red

ıen

ual

the

his

he

.ps :ly

ce

ık

 $w_i = budget share of the i<sup>th</sup> commodity$ 

 $p_i = price of the i<sup>th</sup> commodity,$ 

X = total expenditure on all the commodities.

$$\log P^* = Stone \ s \ index = \sum_{j=1}^n w_j \ \log p_j$$
.

 $D^h = 1$  for regime 'h'

= 0 otherwise

 $F_j^h = p_j$  price of the j<sup>th</sup> commodity in regime 'h'

$$= 0$$
 otherwise

1 22 1

$$G^{h} = \left(\frac{Y}{P^{*}}\right)$$
 'real' total expenditure for regime 'h'

= 0 otherwise

For the full rank regime, the estimated coefficient (except regime one) is obtained by

<sup>&</sup>lt;sup>9</sup> This choice of the functional form has been found to be preferred over the LES. Also the linear approximation of the AIDS model (as in Deaton and Muellbauer, 1980) has been empirically justified. For details on this see Viswanathan (1998).

the sum of the coefficient in the first regime with the corresponding dummy coefficient in that regime as shown in equation A.7a.

perio

hypo

the 1

breal

peric

(that

like

poir

sub<sup>i</sup>

poi

per the

COC

COL

du

ch

5.

5.

T

r

1

ş

## 4.3. Tests for model stability and modelling the dynamic structure

From Table 1 it is noticed that for the given sample there could be several break points (different for different ordinal group) with some of the sub-samples being undersized to allow the estimation of coefficients. The *generalised Chow-test* as in Dufour (1982) allows 'testing of the equality of coefficient vectors in several regressions when the design matrices have arbitrary ranks'. For the samples with sufficient size, the test for structural breaks is reduced to a problem of testing the equality of some or all the coefficients in several regimes. However, in this test the interpretation of the null hypothesis for the undersized samples is not very clear.

Cantrell, Burrows and Vuong (1991) (CBV, henceforth) formalise the implicit null hypothesis of the generalised Chow-test. CBV show that this test is equivalent to introducing a dummy that jointly tests the hypothesis of equality of coefficients across the full rank regimes and the equality of the predictions for the rank deficient regimes with the estimator obtained from the *first* full rank regime. For the rank deficient regimes this test indicates whether the observation(s) has (have) the mean value to be different from that obtained from the first sample. Thus, the analysis is performed via the dummy variable approach. The significance of the dummy coefficients automatically produces indications on the commodities which show a regime shift, and the coefficients that differ between any two regimes. The dummy variable approach can be used to test for equality between the subset of coefficients for the sufficient sized samples. The LA-AIDS specification with the homogeneity and symmetry restrictions (of consumer theory) imposed is estimated in the seemingly unrelated regression equations (SURE) framework. The test as in CBV (is for a single equation linear model) for the SURE framework is briefly described in Appendix A.<sup>10</sup>

<sup>&</sup>lt;sup>10</sup> The tests for structural breaks assume that the variance-covariance matrix (only for the full rank regimes) is the same across the regimes, but this may not be true in reality. However, if this is incorporated in the SURE framework in the dummy variable approach the structure of the variance-covariance matrix would become very complicated and also the number of parameters to be estimated would increase but this aspect could be pursued in a theoretical framework.

unmy

veral

being

as in

*i*eral

with

the

the

icit

: to

DSS

les

nt

be

ia

ts

Ì,

e

The test to identify structural breaks is performed as follows. First the sample period from 4<sup>th</sup> to the 25<sup>th</sup> is tested for model stability. If the tests reject the null hypothesis then the sample is partitioned at the point when IHS rounds begin (that is, the 19<sup>th</sup> round). Within each of these partitioned sub-samples, the test for structural breaks is carried out to ensure that there are no more break points. Also, for the same period an alternative break point is chosen which is one round before the IHS rounds (that is, the 18<sup>th</sup> round). Between the two variants the one which maximises the loglikelihood value is chosen as the model for further tests. Then the next set of sample points belonging to a particular regime are included in the model which is again subjected to the tests for structural breaks. This process continues till all the sample points are exhausted, Also within each of the regimes the test for model stability is performed to ascertain if there are any more break points than the ones anticipated in the beginning. This procedure is repeated for two cases: one, for which a subset of coefficients differ across regimes (referred as model B) and one, for which all the coefficients differ across the regimes (referred as model C). The static model (without dummy variables) is referred as model A. The results on model stability and the changes in the coefficients across regimes are reported in the next section.

## 5. Results

#### 5.1. Main Findings

The results for the final set of tests for overall model stability are reported in Table 2. This test is to compare the likelihood values in the static model (or the restricted model as all the coefficients are restricted to be the same across regimes) with the model incorporating the dummy coefficients. The test statistic follows a chi-square distribution. In columns 2 to 4 a significant (at 5% level of significance) likelihood ratio value indicates that the dynamic model is superior to the static model. In columns 6 and 7 a significant likelihood ratio value indicates that the model indicates that the model that allows for a subset of coefficients to change across regimes is rejected in favour of the one where all the coefficients changes.

13

	Variant A Vs. B		Variant /	A Vs. C	Variant B Vs. $C^*$		
	Rural	Urban	Rural	Urban	Rural	Urban	
Group 1	96.29 <sup>*</sup>	78.45 <sup>*</sup>	87.33 <sup>•</sup>	76.84*	8.45	2.02	
Group 2	133.18*	56.55*	121.2	53.33 <sup>*</sup>	12.01	3.21	
Group 3	98.41 <sup>*</sup>	66.43 <sup>•</sup>	88.51*	60.21 <sup>*</sup>	9.90	6.20	
Group 4	88.97*	82.11	74.79 <sup>*</sup>	76.90 <sup>*</sup>	14.17	5.32	
All groups	93.33*	64.09*	83.69*	62.20 <sup>•</sup>	9.63	1.89	

Table 2. Likelihood Ratio Test for Overall Model Stability

Note: (1) Variant A: Static model without any dummy variables; Variant B: Dynamic model allowing intercept and total expenditure coefficients to change across regimes; Variant C: Dynamic model allowing intercept, total expenditure and price coefficients to change across regimes.

ore

rai

(Ji

pc

se

Se

th

la

ir

p

7

t

1

(2) The test statistic follows a  $\chi^2$  distribution.

\*For rural Group-1 Model B allows only the price coefficients to change across regimes.

The results for overall model stability indicates that for all the ordinal groups and all-groups in both the rural and urban sector the static model is rejected. However, with the exception of two ordinal groups in the rural sector all of them indicate that the model allowing the intercept and the total expenditure coefficients to vary across the sub-samples (of sufficient size) performs better than the model that allows all the coefficients to vary across these regimes. This is true for both the rural and the urban sectors. The inference on which of the coefficients vary across regimes in a particular commodity equation can be had only for the sufficient sized regime, as the coefficients cannot be estimated for the undersized samples. For the latter case, a significant dummy coefficient is interpreted as the deviations from the mean corresponding to the first regime (see equation A.7b of Appendix A). The following table indicates the rank deficient regimes.

	Rural	Urban
Group 1	32, and 38	28, 32, and 38
Group 2	27 to 28, 32, and 38	32, 38, and 42
Group 3	28, 32, and 38	27 to 28, 32, and 38
Group 4	28, 32, and 38	28, 32, and 38
All-groups	28, 32, and 38	28, 32, and 38

Table 3. R	tank Defic	ient Rounds	for Different	Ordinal O	<b>Broups</b>

Note: These regimes are identified based on the predictive test as explained in Appendix A and the detailed results are given in the Tables B.1 in Appendix B.

As noted in Table 3, the rank deficient regimes are not uniform across the ordinal groups for any given sector and also that for some of the ordinal groups the rank deficient regimes are not the same as chosen in the beginning. However, the  $32^{nd}$  (Jul.'77 - Jun.'78) and the  $38^{th}$  (Jan.'83 - Dec.'83) rounds uniformly appear as break points. Also, with the exception of group 1 in the rural sector and group 2 in the urban sector, all the other ordinal groups and all-groups show either the  $27^{th}$  round (Oct.'72 - Sept.'73) or the  $28^{th}$  (Oct.'73 - Jun.'74) round as a break point. As noted before, each of these rounds is a separate regime which could be either due to the data gaps or the larger samples compared to the other rounds. In the absence of information for the intermediate rounds between the quinquennial surveys it is difficult to attribute a particular reason for the occurrence of such departures.

For the full rank regimes the estimates of the coefficients are given in Tables B.2 and B.3 of Appendix B for the rural and the urban sector respectively and the results are discussed below.

The results indicate three regimes for all the ordinal groups and all-groups in both the sectors with the exception of group 1 in the rural sector and groups 1 and 2 in the urban sector. However there are differences between the ordinal groups and between the two sectors with respect to the regime length and the coefficients that vary across the regimes.

## Group 1

In the rural sector, the first full rank regime covers the rounds from 4 to 28 and in the urban sector it covers the rounds 4 to 27. Though for both the rural and the urban sectors the second regime is from 42 to 47, there are differences in the coefficients that change. In the rural sector only the price responses change over the two periods but for the urban sector the dummy coefficient for the intercept and the total expenditure show a change. For the rural sector the model without accounting for the breaks show that 'food' and 'other food' are complements to each other whereas after the dynamic structure is incorporated these groups appear as substitutes in both the regimes. The 'non food' and 'other food' groups are complements in regime 1 (as in the model without structural breaks) but is a substitute in regime 2. The 'food' and the 'non food' groups remain as substitutes with or without dynamic structure. For the rural sector, this could mean that with increases in real income there is perhaps a diversification in the commodity basket and hence the prices have a larger role to play in the decision making process.

## Group 2

For group 2 in both the rural and urban sectors, though the price responses are significant there are no changes across the regimes. The intercept, the total expenditure coefficients and the dummy coefficients attached to them explain most of the variation in consumption for the three-commodity grouping. For the rural sector the first full rank regime covers the rounds 4 to 17; the second full rank regime covers the rounds 4 to 17; the second full rank regime covers the rounds 42 to 47. For the first regime, the 'food' and 'other food' groups are necessary goods and the 'non food' is a luxury good.<sup>11</sup> In regime 2 the dummy coefficients for the intercept and the total expenditure is significant for the 'food' and the 'non food' is significant for the 'food' and the 'non food' is significantly different from regime 1 and unlike regime 1 the 'non food' group appears

as i

and

CON

reç

reç

'n(

ex.

co

an

the

go

in

 $G_i$ 

es

re

si

th

e۶

fc

ťĿ

th

 $\mathbf{b}_{i}$ 

π

d

tl

<sup>&</sup>lt;sup>11</sup> In the LA-AIDS model a negative sign for the total expenditure coefficient in a particular commodity equation indicates the good to be a necessity and a positive sign indicates the good be a luxury item.

as a necessity, indicating the increased importance of this group in the budget set.

For the urban sector, the results are similar to group 1. There are two regimes and the regimes differ in the intercept and total expenditure coefficients. Regime 1 covers the rounds 4 to 16, regime 2, 17 to 28 and regime 3, 43 to 47. For the second regime the dummy coefficient for the total expenditure is different from the previous regime with a significant value in the 'other food' (negative dummy coefficient) and 'non food' (positive dummy coefficient) equations. This has resulted in the expenditure elasticity of 'other food' to be necessity. For the third regime, the dummy coefficients for the intercept and the total expenditure are significant for the 'food' and the 'other food' groups, with a positive sign for the former and a negative sign for the latter commodity group. For the urban sector it is the 'other food' group (luxury good in the first regime and a necessity in the third regime) that assumes importance in the budget set.

#### Group 3

and

the

the

the

the

for

eas

oth

(as

ind

the

+ a

ay

re

al

۶f

)T

e

For the rural sector, in the second regime (rounds 17 to 27) the 'food' group is estimated to be a luxury good compared to it being a necessary good in the first regime (rounds 4 to 16). This is a counter-intuitive result and occurs due to a large and significant positive dummy coefficient for total expenditure in the 'food' equation. In the third regime, covering the rounds 42 to 47, the dummy coefficient for the total expenditure is significant for the 'food' group (with a negative sign) and for the 'non food' group (with a positive sign). This result for the total expenditure shows that though the 'food' and the 'other food' groups are necessities as in the first regime, their magnitudes are much lesser showing that they are more (income) elastic than before. Similarly, the 'non food' group remains to be a luxury good but with a lesser magnitude than in regime 1.

In the urban sector, for regime 2 (rounds 19 to 25), the results show that the dummy coefficients are significant for the intercept and total expenditure with a positive for the 'other food' group and a negative sign for the 'non food' group. For regime 3 (rounds 42 to 47) dummy coefficients for the intercept are significant for all the commodity groups but the dummy coefficients for the total expenditure are

significant for the 'food' (positive sign) and the 'non food' groups (negative sign).

## Group 4

For the rural sector, there are three regimes with respect to the intercept and total expenditure coefficients; rounds 4 to 16 belong to the first regime, rounds 17 to 27 belong to the second and rounds 42 to 47 to the third. The dummy coefficients for the intercept are significant for the 'food' group with a positive sign and the 'non food' group with a negative sign. The dummy coefficients for total expenditure are significant for all the three commodity groups with a positive sign for the 'food' group and with a negative sign for the 'other food' (though at 10 per cent level of significance) and 'non food' groups. In regime 3 covering the rounds 42 to 47, the dummy coefficients for the intercept is significant only for the 'food' group. The price coefficients do not show any change during the period 17 to 27 but for the rounds 42 to 47 there are significant changes in the dummy coefficient for the own prices of 'food', cross price of 'food' with 'other food' and 'non food' groups. With regard to the price coefficients, regime 1 covers the rounds 4 to the 27 and regime 2, 42 to 47.

For the urban sector, the rounds 17 to 27 form regime 2. The dummy coefficients for the intercept is significant (with a positive sign) only for the 'other food' group; and the dummy coefficients for the total expenditure is significant with a positive sign for the 'other food' group and with a negative sign for the 'non food' group. Thus, the 'other food' group is a luxury good for this regime whereas it is a necessity in regime 1; the 'non food' group is a necessity good for this regime whereas it is a luxury in regime 1. The rounds belonging to the third regime are 42 to 47. The dummy coefficient for the intercept is significant only for the 'food' group but the price response for the 'food' group changes for the third regime.

#### All groups

For the rural sector, the price coefficients show no change across any of the regimes. For regime two covering the rounds 17 to 27, the intercept dummy is

inc

'no

anc

to

ne

're

(n)

5.

(1

duir

(nef

significant in the 'food' (positive sign) and the 'non food' (negative sign) groups and so is the case with the 'real' expenditure coefficient. When compared to regime 1 the expenditure elasticity for 'food' increases and that for 'non food' group decreases. With a positive value for the dummy coefficient added to the negative coefficient value in the first regime, we obtain such a result and similarly for the 'non food' group. Regime 3 covers the rounds 42 to 47 with the intercept being significant for all the commodity groups and having a negative value for the 'food' and the 'other food' groups and a positive value for the 'non food' group. This indicates an increase in the average budget share towards the 'non food' group, away from the 'food' and 'other food' groups. The dummy coefficient for 'real' expenditure is significant and positive in the 'food' and the 'non food' groups but is significant and negative for the 'other food' group. The expenditure elasticity in this case does not alter much for the 'food' group but the 'other food' group is less elastic than before.

For the urban sector, regime 2 covers the rounds 17 to 27 with the intercept dummy being significant in the 'other food' (positive sign) and the 'non food' (negative sign) equations. The 'real' expenditure coefficient is significant for all the commodity groups with a negative sign for the 'non food' group and a positive sign otherwise. Further, the expenditure elasticity for the 'food' and the 'other food' groups increases in magnitude compared to regime 1 whereas this value decreases for the 'non food' group. However, the 'food' and 'other food' groups are both necessities and the 'non food' group a luxury in both the regimes. Regime 3 covers the rounds 42 to 47. The intercept dummies are significant for all the commodity groups but has a negative sign for the 'food' and the 'non food' groups. The dummy coefficient for 'real' expenditure is significant for the 'food' (positive sign) and the 'other food' (negative sign) groups.

## 5.2. Discussion of Results

#### (1) The IHS rounds:

and

7 to

for

nồn

are

эир

of

the

s a

ıp. he

٧n

th

2,

ί¥

r

а

• It is noticed that for the regime coinciding with the IHS rounds (19 to 25) or for the regime that includes the IHS rounds (17 to 27), there is a positive sign for the dummy coefficient for the 'food' group and negative sign for the

dummy coefficient for the 'non food' group. Sometimes positive change is also observed for the 'other food' groups.

(2) T

e

For 1

does

has

rour

Bha

regi

(3)

• These changes mentioned are restricted only to the upper expenditure classes and only to the intercept and total expenditure coefficients.

This result is due to the sudden upward (downward) shift in budget shares of the 'food' ('non food') group, increasing the relative importance of the 'food' group in the commodity basket as noted in Viswanathan (1998). During the IHS rounds an integrated schedule was used to gather information on all the socio-economic aspects of a household, namely, demography, employment and unemployment, consumer expenditure and enterprise (Dandekar, 1996 Vaidyanathan, 1986 and Mukherjee, 1986).<sup>12</sup> This resulted in the NSSO collecting both the total expenditure and 'income' together which may have led to some distortions in the responses by the households. Due to this perhaps there was under-reporting of income and hence the total consumption by the respondents. The scope for under-reporting income and expenditure by under-stating 'food' consumption is quite limited because 'food' is a necessary good. The nature of items in the 'miscellaneous non food' category (like durable goods and consumer services), on the other hand, is such that the expenditure on these items could be consistently under reported along with total income and expenditure. As a result, the proportion of 'food' consumption might have got inflated due to the scaling down of the 'non food' expenditure and hence the total expenditure.<sup>13</sup> This may explain why the observed changes in the dummy coefficients is confined to upper ordinal groups because the shares of durables and consumer services in their total expenditure is relatively larger compared to that of the poorer groups. Further, as the distortion is due to the under reporting of total expenditure the econometric results do not show changes in the price coefficients during this period.

20

<sup>&</sup>lt;sup>12</sup> In the surveys prior to this period the schedules relating to different socio-economic aspects were canvassed in relation to different sets of samples. The IHS was discontinued from the 26th round (July 1971 to Sept. 1972).

<sup>&</sup>lt;sup>13</sup>The budget shares add up to one. So a decrease in the budget share of one commodity would be offset by the increase in budget share of some other commodity.

(2) The quinquennial rounds: The data gaps due to the quinquennial surveys show up as break points in the time-series for all the expenditure groups in both the sectors except rural group 1.

For the quinquennial rounds, the absence of a continuous time series expenditure data does not enable us to study the consumer behaviour during this period. Moreover, it has also not been possible to study the effect of a larger sample size during these rounds, an issue that has been discussed in studies related to poverty (Ray and Bhattacharya, 1992). As the estimates of the coefficients cannot be obtained for these regimes it is not possible to explore the underlying reasons for the changes to occur.

(3) After the  $38^{th}$  round:

- For the period beginning from the 42<sup>nd</sup> round, there are changes for the lowest ordinal group in both the rural and the urban sectors. In the rural sector the changes in consumption pattern is reflected as a change in the price responses but in the urban sector this is reflected as a change in the total expenditure coefficients.
- For groups 2 and 3 in the urban sector and group 2 in rural sector the dummy coefficients are positive (significant) for the intercept and total expenditure for the 'food' and 'other food' groups. This may be due to the shift away from cereal to non-cereal food items like meat, egg and fish, and milk and milk products within the 'food' group as noted by Suryanarayana (1995) and Meenakhsi (1996a). These commodities are aggregated into 'food' group in this analysis which may be the reason for a positive dummy coefficient. Also the positive dummy coefficient for the 'other food' group may be due to a significant shift in the expenditure proportions towards vegetables, etc. (included in this group) as noted in the studies mentioned above.
- For group 4 in both the sectors the price responses have changed along with the intercept and total expenditure coefficients. The shift is away from 'food' towards 'non food' and is reflected by a negative dummy coefficient for the intercept and total expenditure for the 'food' group and a positive dummy coefficient for the 'non food' group.
- For all-groups the response is mixed. The intercept and total expenditure coefficients move in opposite directions in both the sectors. For the 'food' group the dummy coefficient is negative for the intercept but positive for the total expenditure variable. This perhaps indicates that though the preferences have

e is

ses

the

the

an

cts
ner
ee,
ne'
ds.
tal
nd
a
ke

re 1d

эđ

al

ts

er

er

ie

e

shifted away from the 'food' group there could be changes in the composition within the 'food' group due to the income effect. Also, for the rural sector the 'non food' dummy coefficient for both the intercept and total expenditure is positive indicating that shift towards the 'non food' group could be both due to changes in preferences and due to the income effect.

be a

ader

are

pre

der

foc

pat

pre

caj

sh

fu

ca

fu

ge

li

b

b

F

v

ءً (

(

• Another important finding is that when the structural breaks are accounted for using the dummy variables, the curvature conditions are satisfied for all the points in the sample space except the rank deficient regimes where no such information could be obtained. This is true for both the rural and urban consumers.

This perhaps indicates that in the demand system without structural changes, the curvature condition was not satisfied because of dynamic mispecification. When the coefficients are estimated the curvature condition was satisfied at all the sample points.

For the post-QS3 rounds the results are similar to the other studies using different methodologies. The study by RR is comparable with the present one as it addresses the issue of changes in consumption pattern based on a (hierarchic) demand system using the dummy variable approach and covers a long enough time period. RR is probably the first attempt to examine the question of changes in tastes within a theoretically and methodologically rigorous framework and is a significant contribution to the consumer behaviour analysis in India. However, the same issue is addressed here in a much larger perspective by improving the methodology that takes into account the following factors.

For India the consumption pattern for the majority of the population in the initial years was at the subsistence level of consumption. In such a case the only constraint binding a consumer's decision would be his income level. But with improvements in the living standards, prices would also appear in a big way. In such a case not only the price variables would appear significant in the econometric model for consumption demand but also substitution possibilities may arise between commodities. The former would rule out Engel curve estimations, requiring a demand system framework and the latter would mean that the functional form specification used for the estimation of the demand system should be effective in modelling this aspect. Therefore, the choice of the LES with additively separable preferences may not

22

be an appropriate specification as the effect of relative price changes are not measured adequately. The LA-AIDS model being a flexible functional form, the price responses are measured effectively, distinguishing it from the income responses, and without any pre-conditions on the parameters prior to the estimation.

n

n

e

n

ž

Also, with the process of development there would be changes in consumption demand due to the Engel's law (over time), that is, a decrease in the budget shares of food and an increase in the budget shares of non food.<sup>14</sup> Such changes in consumption pattern are a natural occurrence and cannot be interpreted as due to changes in preferences. However the LES with constant marginal budget shares would not permit capturing the changes in demand due to Engel's law and the use of this model may show dynamic effects due to mispecification. Therefore, the specification of the functional form should be such that these changes would be precluded and other causes of structural breaks could be identified and modelled. Again the LA-AIDS functional form overcomes this problem as it is derived from the price independent generalised logarithmic cost function that allows for non-linear Engel curves.

The dummy variable approach is appropriate to capture regime shifts in a linear (in parameters) model. It clearly brings out the differences in coefficients between two regimes and permits identification of the regressors that account for such breaks. Further in a linear model, a significant dummy for the constant term would perhaps suggest a change in taste and significant dummies for prices and income would indicate that the changes in demand might have occurred due to some abrupt changes in these variables. With a non-linear model like the LES, such interpretations of dummy coefficients may not be possible.

Finally the break points are tested for their validity unlike in RR where 1973-74 was a pre-determined one. Also, the present study shows that there could be other causes of breaks induced by extraneous factors like changes in methodology of data collection which indirectly affect the independent variables. This is true particularly for the study by RR where the data set includes the rounds belonging to the integrated

<sup>&</sup>lt;sup>14</sup> With increase in incomes the consumers would also prefer to use better quality of the same cereal but such issues are not addressed in any of the studies due to the lack of adequate information.

household survey unlike the other studies mentioned above where the analysis is for the period beginning from 1972-73.

The results here are different from RR: there are multiple break points in the consumption for all the ordinal groups and all-India for both rural and urban consumers; the independent variables that contribute towards the break and the commodities which show breaks in consumption demand vary between the four ordinal groups; and the breaks cannot be attributed to gradual changes in tastes in the conventional sense.

## 6. Conclusion

The present study improves upon the econometric modelling for testing and incorporating structural breaks for an analysis of consumption patterns and also explores the causes of such breaks in the Indian context. In this study it has not been possible to analyse the changes in cereal composition – decline in the budget share of coarse cereals and an increase in the budget share of rice and wheat – an important issue for a developing country. The break-up of the expenditure data on 'foodgrains' into 'cereals' and 'cereal substitutes' is not reported for all the NSS rounds from the beginning. Thus with the data gaps being more frequent and not matching with the gaps in the aggregate 'foodgrains' series, it has not been possible to address this issue in detail at a further level of disaggregation for the corresponding rounds, in an econometric framework. Also, given the small sample covering 32 rounds of the NSS a larger commodity disaggregation would lead to a degrees of freedom problem in order to estimate a demand model like the AIDS.

The results of this study imply that for policy models requiring demand forecasting, the pooling of the data set over the NSS rounds would be inappropriate. The results for the lowest ordinal group in the rural as well as the urban sectors show a decline in the budget shares of food after the 27<sup>th</sup> round (after taking into account the substitution effect). This perhaps is an indication that the calculation of poverty measures should be based on a wider basket of consumption rather than on food-energy-intake method alone because with changing consumption patterns these measures may not adequately capture the living standards of the population

(Pers) const their mode disag set (1

> dyni: info to n pric may mo

(Perspective Planning Division, 1993). The results highlight the differences in consumption patterns of the different economic sections of the population and hence their differences in responses to 'income' and price changes. Therefore, for the policy models that address redistributive issues the consumption model based on disaggregate (forming either decile groups or ordinal classes) rather than a pooled data set (taking all expenditure classes together) would be an effective one.

for

the

ban

the

our

the

nd

so

en

of

nt s'

ıe

ie

le

n S

n

-

100

This study as it is based on all India data is only indicative of the causes of dynamic structure in the consumption patterns. A state level analysis would be more informative to understand the underlying process and the changes over time. Further to measure the price responses adequately, it would be appropriate to use the retail prices. However, such a series is available for a shorter period of time. Therefore one may attempt to measure the price responses and address related issues by estimating a model using the regional level data on consumer expenditures. This extension would also take care of the limitation posed by the reduced sample size.

#### REFERENCES

Mu

Nat

Per

Rac

Ra

Ra

R٤

Sι

- Ahluwalia, M.S., (1978), 'Rural Poverty and Agricultural Performance in India,' Journal of Development Studies, Vol.14, pp. 298-323.
- Brown, J.A.C., and A.S.Deaton, (1972), 'Models of Consumer Behaviour: A Survey', The Economic Journal, Vol.82, 1145-1236.
- Cantrell,R.S., P.M.Burrows and Q.Vuong, (1991), 'Interpretation and Use of Generalised Chow Tests', International Economic Review, Vol.32, No.3, 725-41.
- Chandolk,H.L., (1978), Wholesale Price Statistics in India: 1947-1978, Vol. II, Economic and Scientific Research Foundation House, New Delhi.
- Dandekar, V.M., (1996), The Indian Economy: 1947-92: Vol-2 Population, Poverty and Employment, Sage Publication, New Delhi.
- Deaton, A. and J.Muellbauer, (1980), 'An Almost Ideal Demand System', American Economic Review, Vol.70, 313-24.
- Dufour, Jean-Marie, (1982), 'Generalised Chow Tests for Structural Change: A Coordinate-Free Approach', International Economic Review, Vol.23, 565-75.
- Ghose, S., and N.Bhattacharya, (1994), 'Effect of Reference Period on Engel Elasticities of Clothings and Other Items', *Sarvekshana*, 35-39.
- Government of India, (1979), Report of the Task Force on Projections of Minimum Needs and Effective Consumption Demand, Perspective Planning Division, Planning Commission, New Delhi.

\_\_\_\_, (1987), Index Numbers of Wholesale prices in India: A Time Series Presentation 1971-1986, Office of the Economic Adviser, Ministry of Industries, New Delhi.

- Jain, L.R. and B.S.Minhas, (1991) 'Rural and Urban Consumer Price Indices by Commodity Groups', *Sarvekshna*, Vol. XV, No.1, July-September 1991.
- Majumder, A., (1986), 'Consumer Expenditure Pattern in India : A Comparison of the AIDS and the LES', *Sankhya*, Vol.48, Series-B, Pt.1, 115-43.
- Meenakshi, J.V., (1996a), 'Food Consumption Trends in India: Towards a Better Quality of Diet', Journal of Indian School of Political Economy, July-September, 533-50.

\_\_\_\_\_, (1996b), 'How Important are Changes in Tastes: A State Level Analysis of Food Demand", *Economic and Political Weekly of India*, Vol.XXXI, No.50, 3265-69.

- Minhas, B.S., (1988), 'Validation of Large Scale Sample Survey Data: Case of NSS Estimates of Household Consumption Expenditure', Sankhya, Vo.150, Series-B, Pt.3, 279-326.
- Mukherjee, M., (1986), 'Statistical Information on Final Consumption in India and the National Sample Survey', *Economic and Political Weekly*, Vol.21, No.5, 206-9

26

- Murthy, M.N., and R.Ray, (1989) 'A Computational Procedure for Calculating Optimal Commodity Taxes with Illustrative Evidence from Indian Budget Data', Scandinavian Journal of Economics, 19, 655-70.
- Narayana, N.S., K.S.Parikh, and T.N.Srinivasan, (1991), Agriculture, Growth and Redistribution of Income: Policy Analysis with a General Equilibrium Model of India, North Holland/Allied Publishers, New Delhi.
- Perspectives Planning Division, (1983), Report of the Expert Group on Estimation of Proportion and Number of Poor, Planning Commission, New Delhi.
- Radhakrishna, R., (1991), 'Food and Nutrition: Challenges for Policy', Journal of the Indian Society Of Agricultural Statistics, Vol. XLIII, No.3, 211-27.
  - and K.N.Murthy, (1997), 'Econometrics of Complete Demand Systems with Illustrative Applications for India', in K.L.Krishna (ed.) *Econometric Applications in India*, Oxford University Press, Delhi.
  - and C.Ravi, (1990), 'Food Demand Projection For India', Centre For Economic and Social Studies, Background Paper prepared for 1991 India Country Economic Memorandum.
- and C.Ravi, (1992), 'Effects of Growth, Relative Price and Preference on Food and Nutrition', *Indian Economic Review*, Special Number, 303-23.
- Ray, R., (1986), 'Sensitivity of 'Optimal' Commodity Tax Rates to Alternative Functional Forms: An Econometric Case Study of India', *Journal of Public Economics*, 31, 253-68.
- Ray, S.N. and M.G.Bhattacharya, (1992), 'An Appraisal of the Methodologies and Data Issues Relating to Poverty Analysis', in G.K.Kadekodi and G.V.S.N.Murthy (eds): *Poverty in India: Data Base Issues*, Vikas Publishing House, New Delhi.
- Ravallion, M., and G.Datt, (1996), 'India's Checkered History in Fight Against Poverty: Are there Lessons for the Future?', *Economic and Political Weekly*, Special Number, September 1996, 2479-85.
- Suryanarayana, M.H., (1991), 'A Note on Lorenz Ratios Based on NSS Consumption Distributions', *Journal of Quantitative Economics*, Vol.7, No.1, 179-83.

\_\_\_\_\_, (1995), 'Growth, Poverty and Levels of Living: Hypothesis, Methods and Policies', *Journal of Indian School of Political Economy*, Vol.VII, No.2, 203-55.

\_\_\_\_\_, (1996), 'Poverty Estimates and Indicators: Importance of Data Base', *Economic and Political Weekly*, Vol.XXXI,No.35, 36 & 37, Special number, 1996.

and S.Geetha, (1996), 'P- $\alpha$  Poverty Measure: An Estimable Approach', *Arthavijnana*, Vol.38, 233-47.

Tendulkar, S.D., and L.R.Jain, (1993), 'An Analysis of Inter-State and Inter- Commodity Group Rural-Urban Consumer Price Indices in India, 1983 to 1988-89', Journal of Indian School of Political Economy, Vol.V, No.2.	
<ul> <li>and, (1995), 'Economic Growth and Equity: India, 1970-'71 to 1988- '89', Indain Economic Review, Vol.30, 19-49.</li> <li>, K.Sundaram, and L.R.Jain, (1993), Poverty in India: 1970-'71 to 1988-'89, International Labour Organisation, Asian Regional Team for Employment Promotion (ARTEP), New Delhi.</li> <li>Vaidyanathan,A., (1986), 'On the Validity of NSS Consumption Data', Economic and Political Weekly, Vol.XXI, Vol.3, 129-37.</li> <li>Viswanathan,B., (1998), Some Econometric Issues in Consumer Behaviour Analysis, Unpublished Ph.D. Thesis Submitted to Indira Gandhi Institute of Development Research, Mumbai.</li> </ul>	parai dema obse (Zell $Y_{0(F)}$ The
Development Research, Munidal.	E(1
	whe as: E (
	$E_{L}$ $\Sigma_{C}$ $I_{T}$
	$\hat{\boldsymbol{\beta}}_{\boldsymbol{\beta}}_{\boldsymbol{\beta}}$
	is, rej
	Y It su
	th ec

#### APPENDIX A

Let there be P commodities (or cross-sections) each with T observations and K parameters (regressors) to be estimated in the demand model. For the purpose of demand system estimation suppose each of these P commodity equations (with T observations each) represent each of the (n-1) commodities. Then the SURE model (Zellner, 1962) is commonly represented as follows:

$$Y_{0(PT\times I)} = X_{0(PT\times PK)}\beta_{0(PK\times I)} + u_{(PT\times I)}$$
A.1

The vector of disturbances has a covariance matrix:

$$E(uu') = \begin{bmatrix} \sigma_{11}I_T & \sigma_{12}I_T & \cdots & \sigma_{1P}I_T \\ \sigma_{21}I_T & \sigma_{22}I_T & \cdots & \sigma_{21}I_T \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ \sigma_{1P}I_T & \sigma_{2P}I_T & \cdots & \sigma_{PP}I_T \end{bmatrix} = \Sigma_0 \otimes I = \Gamma_{PT \times PT}$$
 A.2

where, the correlations between the disturbances in different equations is expressed as:

$$E(u_{ii}u_{js}) = \sigma_{ij} \quad t = s$$
  

$$E(u_{ij}u_{js}) = 0 \quad t \neq s \quad for \ i, j = 1, 2, \dots, P$$

 $\Sigma_0$  = the variance-covariance matrix of contemporaneous correlation and  $I_T$  = identity matrix of order T.

The estimator of the coefficient in the SURE model is given by

$$\hat{\beta}_0 = (X'_0 \Gamma^{-1} X_0) (X'_0 \Gamma^{-1} Y_0)$$
 A.3

Suppose the hypothesised structural change occurs at L-1 different points that is, there are L regimes, with  $T_h$  observations in each of them. This model is represented as:

$$Y^{*(h)} = X^{*(h)} \beta^{*(h)} + e^{*(h)}, \quad h = 1, 2, \dots, L$$
 A.4

It is possible that some of the regimes may have a rank deficient design matrix. In such regimes it is not possible to estimate the  $\beta$  coefficients. If  $X^{*(h)}$  is of full rank then the corresponding  $\beta^{*(h)}$  can be estimated as a SURE model for each of the regime as in equation A.3.

The L regimes could be stacked to obtain a single model and represented as:

For

 $\hat{\beta}^{(1)}$ 

for

 $\hat{\alpha}^{(i)}$ 

Th

lik

res

wu foi

to

co

re

ço

$$Y^* = X^* \beta^* + e \tag{A.5}$$

where,

$$\boldsymbol{X}^{\star} = \begin{bmatrix} \boldsymbol{X}^{(1)} & & & \\ & \boldsymbol{X}^{(2)} & & \\ & & \ddots & \\ & & & \ddots & \\ & & & \boldsymbol{X}^{(L)} \end{bmatrix}_{\boldsymbol{PT} \times \boldsymbol{LPK}} \text{ and } \boldsymbol{\beta}^{\star} = \begin{pmatrix} \boldsymbol{\beta}^{(1)} \\ \boldsymbol{\beta}^{(2)} \\ \vdots \\ \boldsymbol{\beta}^{(L)} \end{pmatrix}_{\boldsymbol{LPK} \times \boldsymbol{I}}$$

The model in equation A.5 is referred to as *unrestricted model* and the model in equation A.1 is referred to as the *restricted model*. In order to test for the model stability, identify the location of breaks and explanatory variables contributing towards these breaks the methodology suggested by Cantrell *et al.*, (1991) is used. In this a likelihood-ratio test is constructed that jointly tests the hypothesis of equality of  $\beta$  coefficients across the full rank regimes and the equality of the predictions for the rank deficient regimes with the estimate obtained from the *first* regime (which is essentially of full rank).

Without loss of generality, we can stack the full column rank matrices together and keep the rank deficient ones below. Then the model can be stacked as:

$$\begin{bmatrix} Y_{l} \\ Y_{2} \\ \vdots \\ X^{(2)} & X^{(2)} \\ \vdots \\ X^{(2)} & X^{(2)} \\ \vdots \\ \vdots \\ Y_{q} \end{bmatrix} = \begin{bmatrix} X^{(1)} & 0 & \vdots \\ X^{(2)} & X^{(2)} \\ \vdots \\ X^{(q)} & 0 & \vdots \\ X^{(q)} & 0 & \vdots \\ X^{(q)} & 0 & I_{T_{r''}} \\ \vdots \\ \vdots \\ X^{(l)} & 0 & \vdots \\ X^{(l)}$$

For the full rank regimes the SURE estimators is obtained as in equation A.3:

$$\hat{\beta}^{(1)}, \quad \hat{\alpha}^{(l)} = (\hat{\beta}^{(l)} - \hat{\beta}^{(1)}) \text{ for } i = 2,..., q \text{ and}$$
 A.7a

for the rank deficient regimes:

$$\hat{\alpha}^{(l)} = (Y^{(l)} - X^{(l)} \hat{\beta}^{(1)}) \text{ for } i = q+1,...,L.$$
 A.7b

Therefore, the joint test is to test the nullity of  $\alpha^{(2)}, ..., \alpha^{(L)}$ . In order to do this a likelihood ratio test comparing the unrestricted model in equation A.6 with the restricted model in equation A.1 is used. The rejection of the joint null hypothesis would imply that the coefficients are not stable over the entire sample period. Further, for each of the full rank regimes the significance of the dummy coefficients attached to the different regressors give information on whether the coefficients for a particular commodity group differs from the first regime or not. For each of the rank deficient regimes to the first regime or not.

## APPENDIX B

		Rural			Urban	
Group 1	32 <sup>nd</sup>	38 <sup>th</sup>	aanaan maaniga maana maada an mada an ah	28 <sup>th</sup>	32 <sup>nd</sup>	38 <sup>th</sup>
Food	-0.032**	-0.071*		0.011	-0.028	-0,086
	(-1.74)	(-3.65)		(0.6)	(-1.1)	(-4,4)
Other Food	0.039*	0.045*		0.047*	0.021	0.050
	(3.35)	(3.21)		(2.64)	(1.87)	(2.69)
Non Food	-0.006	0.026		~0.058	0.007	0.036
	(-0.04)	(0.15)	x	(-0.27)	(2.03)	(0.17)
Group 2	27 to 28 <sup>th</sup>	32 <sup>nd</sup>	38 <sup>th</sup>	32 <sup>nd</sup>	38 <sup>th</sup>	42 <sup>nd</sup>
Food	-0.039*	-0.086*	-0.126*	-0.004	-0.081*	-0.051*
	(-2.01)	(-4.68)	(-7.00)	(-0.21)	(-4.12)	(-2.21)
Other Food	0.031*	0.033*	0.050*	0.043**	0.055*	0.061*
	(2.33)	(2.54)	(3.60)	(1.88)	(2.05)	(2.26)
Non Food	0.008	0.053	0.076	-0.039	0.026	-0.010
	(0.07)	(0.46)	(0.66)	(1.22)	(0.88)	(-0.27)
Group 3	28 <sup>th</sup>	32 <sup>nd</sup>	38 <sup>th</sup>	27 to 28 <sup>th</sup>	32 <sup>nd</sup>	38 <sup>th</sup>
Food	-0.039**	-0.053	-0.125	0.018	-0.015	-0.063*
	(-1.89)	(-3.27)	(-7.79)	(1.00)	(0.94)	(-3.33)
Other Food	-0.005	0.046*	0.048	0.018	0.034*	0.052*
	(-0.28)	(3.27)	(3.41)	(1.15)	(2.56)	(3.23)
Non Food	0.044	0.006	0.078	-0.036	-0.019	-0.010
	(0.46)	(0.07)	(0.79)	(-2.33)	(-0.18)	(0.10)
Group 4	28 <sup>th</sup>	32 <sup>nd</sup>	38 <sup>th</sup>	$28^{th}$	32 <sup>nd</sup>	38 <sup>th</sup>
Food	-0.038*	-0.111*	-0.137*	-0.002	-0.068*	-0.118*
·	(-2.26)	(-5.49)	(-6.43)	(-0.09)	(-3.21)	(-7.02)
Other Food	0.023	-0.013	0.052	0.015	-0.003	0.026
	(0.68)	(-0.67)	(2.31)	(0.046)	(-0.12)	(1.20)
Non Food	0.014	0.124	0.084	-0.013*	0.072*	0.091*
	(0.26)	(2.29)	(1.57)	(-2.44)	(2.44)	(3.09)
All groups	28 <sup>th</sup>	32 <sup>nd</sup>	38 <sup>th</sup>	28 <sup>th</sup>	32 <sup>nd</sup>	38 <sup>th</sup>
Food	-0.026**	-0.078*	-0.121*	-0.006*	-0.035*	-0.088*
	(-1.94)	(-6.02)	(-7.77)	(-2.42)	(-3.24)	(-8.06)
Other Food	0.014	0.021	0.051*	0.012	0.037*	0.054*
	(2.71)	(1.45)	(3.56)	(0.49)	(2.21)	(2.90)
Non Food	0.011	0.057	0.069	-0.006	-0.003	0.035
	(0.09)	(0.43)	(0.52)	(-0.06)	(-0.04)	(0.37)

Table B.1. NSS Round-wise Predictive Test Results to Identify Rank Deficient Regimes

Note: 1. The reported test results are the differences in the predicted values as in equation A.7b of Appendix A.
2. The values in brackets are the t-ratios.

			GRC	OUP I			
		Interc	epts (no cha	nge across re	egimes)	an gan gan a sa s	
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^1$	0.442*	0.282	0.275*				
	(4.5)	(5.54)	(3.00)				
			Price co	efficients			
J	Regime 1 (r	ounds 4 to	28)				
Coeffs.	Food	Othfd	Nonfd				
γli	0.161	n The Annual Market Conference and Conference and Conference and Conference and Conference and Conference and C	an chu shi dharan bir na bi				
	(2.34)						
$\gamma_{2i}^{1}$	-0.107*	0.139*			,		
	(-2.38)	(3.86)					
$\gamma^1_{3i}$	-0.055	-0.032	0.086*				
	(-1.34)	(-1.64)	(2.37)			ádist (mmar a si Milin da caintáití acainteac	
		· ·	Regime 2 (ro	ounds 42 to 4	7)		
D	ummy coef	ficients for	price		Price co	oefficients	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
$d\gamma_{1i}^2$	0.965			$\gamma_{11}^2$	1.126*		
	(2.31)				(2.70)		
$d\gamma_{2i}^2$	-0.563*	0.306*		$\gamma_{2i}^2$	-0.670	0.445	
	(-5.2)	(4.54)			(-6.38)	(7.57)	
$d\gamma_{3i}^2$	-0.402	0.257*	0.145	$\gamma_{3i}^2$	-0.456	0.225	0.231
	(-0.92)	(2.19)	(0.29)		(-1.04)	(0.52)	(0.18)
	Tota	al expenditu	ure coefficien	nts (no chang	ge across re	egimes)	
Coeffs.	Food	Othfd	Nonfd				
β¦	-0.222	0.183	0.039				
-	(-2.58)	(3.53)	(0.35)		×		

NOTE: (1)The Greek letters represent different coefficients for the different regimes as in equation 1. (2)The subscript refers to a particular commodity group and the superscript refers to the particular full rank regime as mentioned in the table.

)

8

6)

;

į

(3) The coefficient for the first regime is represented with the superscript '1' and for the other regimes it is the sum of the dummy coefficient (preceded by a 'd' as shown in equaiton 1) and the corresponding cofficient in the first regime as shown in equation A.7a.

			GRC	OUP 2			
		andere en al se a constant de la constant en anticipa de la constant de la constant de la constant de la const Internet de la constant de la consta	Inte	rcepts		a ya a a yana ya kuta ya ya kuta ya kut	
P	legime 1 (r	ounds 4 to	17)				
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^1$	0.291	0.118*	0.590				
	(3.79)	(2.18)	(10.36)				
Dum	my Coeffic	cients for Ir	itercept		Inte	ercept	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
40.84.95.4 8.5 45.95.94 (Annuel Statement			Regime 2 (ro	ounds 19 to 2	25)		e a se contra e na de particular de la contra e na de particular de particular de particular de particular de p
$d\alpha_i^2$	0.260**	0.106	-0.366*	$\alpha_i^2$	0.552*	0.224*	0.224
•	(1.75)	(0.97)	(-3.18)	1	(4.11)	(2.32)	(2.18)
		*	Regime 3 (ro	ounds 42 to 4	47)		
da <sup>3</sup>	0.091	0.285*	-0.376*	$\alpha_i^3$	0.382*	0.404*	0.214*
	(0.54)	(2.48)	(-3.27*)		(2.72)	(4.22)	(2.18)
		Price Cod	efficients (no	change acro	oss regimes	5)	
Coeffs.	Food	Othfd	Nonfd				
γli	0.101		****				
• 11	(1.4)						
$\gamma^{I}_{2i}$	0.012	-0.010					
	(0.26)	(-0.31)					
$\gamma_{3i}^{1}$	-0.113*	0.002	0.113*				: 4
	(-2.6)	(-0.02)	(3.21)				
		Л	'otal expendi	ture coeffic	ients		
l	Regime 1 (	rounds 4 to	17)				
Coeffs.	Food	Othfd	Nonfd				
β	-0.268*	-0.045	0.313*				
• •	(-4.14)	(-1.00)	(6.52)				
Dumn	y coefficie	ents for tota	al expend.	Тс	tal expend	liture coeffi	cients
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (r	ounds 19 to	25)		
$d\beta_i^2$	0.235*	0.056	-0.291*	$\beta_i^2$	-0.034	0.011	0.022
1 1	(2.07)	(0.67)	(-3.27)	F 1	(-0.34)	(0.16)	(0.29)
			Regime 3 (r	ounds 42 to	47)		
dβi	0.213**	0.148**	-0.361*	$\beta_i^3$	-0.055	0.103	-0.047
1.1	(1.87)	(1.77)	(-3.74*)	r" (	(-0.48)	(1.31)	(-0.58)

34

			GRO	OUP 3			
nano kaj konzero por degendo este decimen	annanian tama magananiki minjasi aping minj	ányán kyulopi laki nanyangai ar man tepana ana ak	Inte	rcepts		andaya ana ang ang ang ang ang ang ang ang an	
R	legime I (r	ounds 4 to	16)				
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^!$	0.447*	0.112*	0.441*				
	(11.25)	(3.29)	(14.34)				
Dum	my Coeffic	cients for In	tercept		Inte	ercept	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
		]	Regime 2 (ro	ounds 17 to 2	27)	**************************************	
$d\alpha_i^2$	0.152	0.054	-0.206*	$\alpha_i^2$	0.600*	0:166**	0.234*
	(1.32)	(0.55)	(-2.11)		(5.44)	(1.81)	(2.49)
	×.		Regime 3 (r	ounds 42 - 4	7)		
$d\alpha_i^3$	0.001	0.083	-0.085	$\alpha_i^3$	0.449*	0.196*	0.356*
,	(0.02)	(1.06)	(-0.87)	•	(4.96)	(2.64)	(4.87)
		Price coe	fficients (no	change acro	oss regimes	5)	
Coeffs.	Food	Othfd	Nonfd				
γli	0.113*	<b>1</b>	98649-18-49-04-89-96-96-96-96-96-97-97-97-97-97-97-97-97-97-97-97-97-97-				
• 11	(2.23)						
$\gamma_{2i}^{I}$	-0.016	0.045					
	(-0.45)	(1.29)					
$\gamma_{3i}^1$	-0.097*	-0.029	0.126*				
	(-3.14)	(-1.2)	(4.45)				
		Т	otal expend	iture coeffic	ients		
]	Regime 1 (	rounds 4 to	16)				
Coeffs.	Food	Othfd	Nonfd				
$\beta_i^1$	-0.137*	-0.083*	0.219*		,		
·	(-3.11)	(-2.16)	(6.37)				
Dun	uny coeffic	cients for to	tal exp.	To	otal expend	liture coeffi	cients
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (1	ounds 17 to	27)		
$d\beta_i^2$	0.169	0.042	-0.212*	$\beta_i^2$	0.032	-0.040	0.008
• •	(1.38)	(0.41)	(-2.01)	••	(0.29)	(-0.43)	(0.08)
			Regime 3 (I	rounds 42 to	47)		
dβi	-1.430	-0.011	0.154**	βi	-0.289	-0.094	0.373
• •	(1.51)	(-0.1)	(-1.77)	••	(0.44)	(-1.02)	(0.49)

		. coefficiei	its for Dynan			aan (Comu	., ) Manananan
anna a'	latitis (1961) - terioan arai an	a). Media di San dipo de la produce de la segunda de la	*****	OUP 4	an an Manufacture and a sub-		
	-	1.748×1.044/0000000000000000000000000000000000		rcepts	ár de tir bindr ill strar flataják illestajer sere	110-148 (110) of an information of a state of the state o	
e		rounds 4 to 1	-				
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^i$	0.419*	0.182*	0.398				
***	(26.04)	(9.74)	(17.19)	anivatoriumiu de Alineirijai (* 1653)			
Du Coeffs.	mmy Coeffi Food	cients for In Othfd	Nonfd	Coeffs.	Food	ercept Othfd	Monfd
Coens.	ruou	Ouna		ounds 17 to 2	******	Ounu	Nonfd
• 2	0.107*	-0.007	-0.010		0.526*	0.174	0.299*
$d\alpha_i^2$	(2.45)			$\alpha_i^2$	(13.22)		(4.94)
	(2.43)	(-0.14)	(-1.53)	ounds 42 to 4		(3.51)	(4.94)
. 1	-0.109*	0.035	0.074		0.310	0.216 *	0 472*
$d\alpha_1^3$				$\alpha_i^3$			0.473
	(-3.79)	(1.01)	(1.14)	oefficients	(12.56)	(7.31)	(12.61)
	Domina 1 (	rounda A to		)emcients			
Coeffs.	Food	rounds 4 to Othfd	27) Nonfd				
	0.059**	Juliu	1401110				
ΥI							
	(1.98)	0.016					
γ <sup>1</sup> <sub>2i</sub>	-0.043	-0.016					
	(-1.11)	(-0.62)					
$\gamma_{3i}^1$	0.078	-0.008	0.051*				
	(3.27)	(-1.11)	(2.69)				
		ficients for p		a		oefficients	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
	,		Regime 2 (ro	ounds 42 to 4			
$d\gamma_{1i}^2$	0.124**			$\gamma_{1i}^2$	0.183*		
	(1.72)				(2.43)		
$d\gamma_{2i}^2$	0.132**	-0.146		$\gamma_{2i}^2$	0.089	-0.067	
- 41	(1.77)	(-1.51)			(1.19)	(-0.802)	1
$d\gamma_{3i}^2$	-0.256**	0.013	0.242	$\gamma_{3i}^2$	-0.272**	-0.021	0.293*
1.1	(-3.65)	(0.10)	(1:54)	131	(-1.95)	(-0.24)	(2.01)
			Total expendi	ture coefficie	nts		
	Regime 1 (	rounds 4 to 1	16)				
Coeffs.	Food	Othfd	Nonfd				
βi	-0.017	0.016**	0.001				
P1	(-2.34)	(199)	(3.02)				
Du		cients for tot	al exp.	T	'otal expend	liture coeffici	ents
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
				ounds 17 to 2			
$d\beta_i^2$	0.229**	-0.101	-0.128	$\beta_i^2$	0.212**	-0.084*	-0.127*
••	(-1,84)	(-1.96)	(-2.67)	• •	(1.94)	(-2.15)	(2.55)
			Regime 3 (ro	ounds 42 to 4	7)		
3	-0.006*	-0.081**	0.087	03	-0.023*	-0.065**	0,088
dβi	0.000	0.001	0.007	βi	0.0.00	0.000	0.000

****	ngginning 70 liningsangan pentahagtar	<del></del>	ALL-C	BROUPS		e 1997-1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 -	1994-1994
Manua distance costre w		Mart of Sectors, And Sectors (Martin Sectors)	Inte	rcepts	agina daamadaa waxaa waxaa daa daa daa da	eoleman en	10
1	Regime 1 (	rounds 4 to	16)				
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^1$	0.399*	0.116*	0.485*				
	(7.33)	(2.27)	(8.60)				
Dum	umy Coeffi	cients for Ir	ntercept		Inte	ercept	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
		<b></b>	Regime 2 (ro	ounds 17 to 2	!7)		
$d\alpha_i^2$	0.058*	0.025	-0.083*	$\alpha_i^2$	0.456*	0.140	0.403*
•	(3.51)	(0.22)	(-2.62)	·	(4.96)	(1.39)	(3.47)
			Regime 3 (ro	ounds 42 to 4	17)		
$d\alpha_i^3$	-0.116*	-0.006**	0.122*	$\alpha_i^3$	0.283	0.109	0.608**
•	(-2.42)	(-1.92)	(3.91)	•	(1.02)	(0.39)	(1.95)
		Price coe	fficients (no	change acro	ss regimes	;)	
Coeffs.	Food	Othfd	Nonfd				
γli	0.104*						
	(2.53)						
$\gamma^1_{2i}$	-0.006	-0.008					
	(-0.18)	(-0.29)					
$\gamma_{3i}^{I}$	-0.098*	0.014	0.074				
•••••	(-2.99)	(0.38)	(2.65				d
		Т	otal expendi	ture coeffici	ents		
I	Regime 1 (	rounds 4 to	16)				
Coeffs.	Food	Othfd	Nonfd			,	
β¦	-0.140*	-0.063	0.203*				
	(-2.40)	(-1.13)	(3.30)				
Dumm	ny coeffici	ents for tota	l expend.	То	tal expend	iture coeffi	cients
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (ro	ounds 17 to 2	27)	¥1)	
dβį	0.067*	0.003	-0.071*	$\beta_i^2$	-0.073	-0.059	0.132
. * 1	(2.59)	(0.02)	(-3.51)	F1	(-0.83)	(-0.61)	(2.18)
			Regime 3 (ro	ounds 42 to	47)		
dβ <sup>3</sup> <sub>i</sub>	0.011	-0.115**	0.104	β <sup>3</sup>	-0.129*	-0.178	0.307*

......

0.000

			GRC	OUP 1			Contraction of the second second second
inalde i pul opport i pula e un pul i pud car		*****	Inter	cepts	4		n de la companya de l
I	Regime 1 (	rounds 4 to	27)	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Wijkermaan van Hillie in de aal van heesteleyn.	n an find rinnen men en personal dag da las desens	1786-1996-1996-1996-1996-1996-1996-1996-19
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^l$	0.371	0.379*	0.250				
-	(3.78)	(4.16)	(1.57)				
Dun	ımy coeffi	cients for in	tercept		Int	ercept	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
		J	Regime 2 (ro	unds 42 to 4			
$d\alpha_i^2$	-0.116*	0.113	0.003	$\alpha_i^2$	0.254*	0.492*	0.254
	(-2.75)	(0.68)	(0.02)	•	(2.08)	(3.74)	(1.15)
		Price coe	fficients (no	change acro	ss regimes	)	
Coeffs.	Food	Othfd	Nonfd				
$\gamma_{1i}^{i}$	0.024*						,
	(2.64)						
$\gamma^1_{2i}$	-0.032*	0.030					
	(-2.21)	(3.18)					
$\gamma_{3i}^{1}$	0.008	0.001	-0.009**				
	(0.23)	(0.04)	(-1.97)				
		Т	otal expendit	ture coeffici	ents		
I	Regime 1 (	rounds 4 to	27)				
Coeffs.	Food	Othfd	Nonfd				
$\beta_i^1$	-0.104	0.105**	-0.001				
	(-1.64)	(1.78)	(-0.01)				
Dun	my coeffic	cients for to	tal exp.	To	tal expend	iture coeffi	cients
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (ro	unds 42 to 4	47)		
$d\beta_i^2$	-0.017	0.033*	-0.016	$\beta_i^2$	-0.122	0.139	-0.017
•	(-4.21)	(0.42)	(-1.98)	* •	(-1.29)	(1.35)	(-0.1)

Table B.3. Coefficients for Dynamic LA-AIDS Model: Urban

NOTE: (1)The Greek letters represent different coefficients for the different regimes as in equation 1. (2)The subscript refers to a particular commodity group and the superscript refers to the particular full rank regime as mentioned in the table.

(3) The coefficient for the first regime is represented with the superscript '1' and for the other regimes it is the sum of the dummy coefficient (preceded by a 'd' as shown in equaiton 1) and the corresponding cofficient in the first regime as shown in equation A.7a.

		and the states of the state of the	GRO	DUP 2		980000-000-000-000-000-000-000-000-000-0	
			Inte	rcepts			
	Regime 1 (	rounds 4 to	25)				
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^1$	0.288*	0.254*	0.457*				
	(5.82)	(4.61)	(5.97)			924-00-554-00-00-00-00-00-00-00-00-00-00-00-00-00	
		1	Regime 2 (ro	ounds 42 to 4	47)		
Dum	uny coeffic	ients for in	tercept		Inte	ercept	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
$d\alpha_i^2$	0.245***	0.360*	-0.606*	$\alpha_i^2$	0.408*	0.506*	0.086
	(1.93)	(2.55)	(-5.14)		(3.82)	(4.24)	(0.55)
		Price coe	fficients (no	change acro	oss regimes	)	
Coeffs.	Food	Othfd	Nonfd				
$\gamma_{1i}^1$	-0.039*		99999				
	(-2.8)						
$\gamma_{2i}^{I}$	-0.003*	0.039*					
	(-3.07)	(2.09)					
$\gamma_{3i}^{l}$	0.042**	-0.037*	-0.005*				
·····	(1.90)	(-3.06)	(-3.09)				
		Т	otal expendi	iture coeffic	ients		
]	Regime 1 (1	rounds 4 to	16)				
Coeffs.	Food	Othfd	Nonfd				
β¦	-0.170*	0.014	0.156*				
, ,	· (-3.98)	(0.28)	(2.30)				
Dun	nmy coeffic	cients for to	otal exp.	Te	otal expend	iture coeff	icients
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (r	ounds 17 to	28)		
$d\beta_i^2$	-0.005	-0.030*	0.035*	$\beta_i^2$	-0.175*	-0.016	0.191*
	(-0.50)	(-2.53)	(2.03)		(-4.02).	(0.27)	(1.85)
			Regime 3 (r	ounds 43 to	47)		
$d\beta_i^3$	0.080*	-0.085*	0.005	$\beta_i^3$	-0.090**	-0.071	0.161

			GR(	OUP 3			
			Inte	rcepts	ni kirologi yang bilan kirologi kana kana da persata kirologi kana kirologi yang bilan kirologi yang bilan kiro	an a	
F	Regime I (r	ounds 4 to	17)				
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^1$	0.219*	0.150*	0.631				
•	(5.06)	(4.31)	(12.32)				
Dum	my coeffic	ients for in	lercepts		Inte	ercepts	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
<u></u>		******	Regime 2 (ro	ounds 19 to 2	25)	an bir cinin katalan katalan da bir kepan yang sana dangan	
$d\alpha_i^2$	-0.057	0.245*	-0.188**	$\alpha_i^2$	0.162**	0.395*	0.443*
	(-0.62)	(3.06)	(-1.74)	• 、	(1.76)	(5.06)	(4.23)
			Regime 3 (ro	ounds 42 to	47)		
$d\alpha_i^3$	0.094**	0.123*	-0.217	$\alpha_i^3$	0.313*	0.273*	0.414*
	(1.91)	(3.04)	(-2.01)		(11.15)	(10.84)	(12.54)
		Price coe	efficients (no	change acro	oss regimes	;)	
Coeffs.	Food	Othfd	Nonfd	·····		arial Erre Chenden Handland Angel Careta ang ang ang	
$\gamma_{1i}^1$	-0.071	*********					
	(-1.35)					•	
$\gamma^{1}_{2i}$	-0.051	0.061**					
	(-1.42)	(1.75)					
$\gamma_{3i}^{1}$	0.122*	-0.010	-0.112*				
	(2.97)	(-0.33)	(-2.32)				
		]	fotal expend	iture coeffic	ients		
]	Regime I (1	rounds 4 t0	17)				
Coeffs.	Food	Othfd	Nonfd				
β¦	-0.269*	-0.155	0.424				
	(-4.39)	(-3.06)	(5.81)				
Dun	nmy coeffic	cients for to	otal exp.	Te	otal expend	liture coeffi	cients
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (r	ounds 19 to	25)		1997 Rose of the second se
$d\beta_i^2$	-0.092	0.299*	-0.207*	$\beta_i^2$	-0.361	0.143	0.217
• •	(-0.74)	(2.75)	(-1.92)	• ;	(-3.08)	(1.43)	(1.61)
			Regime 3 (r	ounds 42 to	47)		
$d\beta_i^3$	0.244**	0.113	-0.357	$\beta_i^3$	-0.025	-0.042	0.068
	(1.97)	(1.04)	(-3.46)		(-0.42)	(-0.73)	(0.91)

1.8

1

Table B.3. Coefficients for Dynamic LA-AIDS Model: Urban (Contd.)

27, <del>27, 18, 18, 1</del> 7, 19,			GR	OUP 4		Dependential and difficult claim	
*****			Inte	srcepts	in The Constraint States and State	1999 1997 1997 1997 1997 1997 1997 1997	an da an in suisin adairte
	Regime 1 (	rounds 4 to		eren in the second s			alalalan yang kanang
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^1$	0.296*	0.259*	0.444*				
~1	(17.53)	(12.03)	(14.74)				
Dun	nmy coeffi	cients for in	itercept	tin vita o starochaine anges mandeta	Int	ercept	anna derbarten derbin Könten anderenge
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (re	ounds 17 to 2	•		
$d\alpha_i^2$	-0.003	0.046**	-0.042	$\alpha_i^2$	0.293*	0.305*	0.402*
	(-0.21)	(1.94)	(-1.23)		(4.86)	(3.06)	(6.12)
			Regime 3 (r	ounds 42 to 4	· .	•	
$d\alpha_i^3$	-0.116	0.113	0.003	$\alpha_i^3$	0.237*	0.295*	0.468*
2-0 <del>1.110711111111111111111111111111111111</del>	(-0.75)	(0.68)	(0.02)	and the second secon	(9.78)	(9.59)	(10.50)
on singly because of the same	energen anternant anter antera procession			oefficients		1	
		rounds 4 to					
Coeffs.	Food	Othfd	Nonfd				
$\gamma_{1i}^1$	-0.083**						
	(-1.78)						
$\gamma_{2i}^{1}$	0.098*	0.005	,				
	(2.22)	(0.08)					
$\gamma_{3i}^1$	-0.014	-0.103	0.117				
	(-0.29)	(-1.54)	(1.30)				
Dı	ummy coef	ficients for	price		Price c	oefficients	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (re	ounds 42 to	,		
$d\gamma_{1i}^2$	0.206**			$\gamma_{1i}^2$	0.122		
	(1.98)				(1.19)	,	
$d\gamma^2_{2i}$	-0.155*	-0.157		$\gamma_{2i}^2$	-0.057	-0.152	
	(-1.84)	(-1.39)			(-0.64)	(-1.48)	
$d\gamma_{3i}^2$	-0.051	0.312	-0.261	$\gamma_{3i}^2$	-0.065	0.209	-0.144
,	(-0.79)	(1.68)	(-1.35)		(-1.66)	(2.51)	(-1.11)
,		Т	otal expendi	ture coeffici	ients		
	Regime 1 (1	rounds 4 to	16)				
Coeffs.	Food	Othfd	Nonfd				
$\beta_i^1$	0.007	-0.151*	0.144**				
	(0.16)	(-2.65)	(1.89)				
	uny coeffic	cients for to	tal exp.	To	-	iture coeffi	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
	0.55		Regime 2 (ro				
$d\beta_i^2$	-0.064	0.156	-0.091	β²	-0.057	0.005	0.051
	(-3.76)	(1.77)	(-0.57)		(-0.79)	(0.06)	(0.37)
	0.017		Regime 3 (ro			A + A +	0.107
$d\beta_i^3$	-0.045	0.051	-0.005	$\beta_i^3$	-0.037	-0.101	0.137
••	(-0.54)	(0.52)	(-0.04)		(-0.55)	(-1.21)	(1.25)

41

			•				
			ALL-G	ROUPS			
			Inter	cepts			
	Regime	1 (rounds 4	to 16)				
Coeffs.	Food	Othfd	Nonfd				
$\alpha_i^1$	0.228*	0.102**	0.670*				
	(7.10)	(1.88)	(10.79)				
Dum	uny coeffic	cients for int	ercept		' Inte	ercept	
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
		· I	Regime 2 (ro	unds 17 to 2	27)		
$d\alpha_i^2$	0.070	0.181*	-0.252*	$\alpha_i^2$	0.299*	0.283*	0.418*
	(1.49)	(2.25)	(-2.64)		(7.81)	(4.28)	(5.30)
		I	Regime 3 (ro	ounds 42 to 4	47)		
$d\alpha_i^3$	-0.029**	0.092*	-0.062*	$\alpha_i^3$	0.199*	0.193*	0.608*
	(-1.97)	(2.86)	(-3.65)		(3.80)	(2.23)	(5.81)
		Price coe	fficients (no	change acro	oss regimes	3)	
Coeffs.	Food	Othfd	Nonfd				
$\gamma_{1i}^{J}$	-0.048						
	(-1.53)						
γ <sup>1</sup> <sub>2i</sub>	0.029	0.021					
	(1.10)	(0.53)					
$\gamma_{3i}^{I}$	0.017*	-0.050	0.033**				
	(3.78)	(-1.35)	(1.74)				
	(U.c.p	T	otal expendi	ture coeffic	ients		
	Regime 1 (	(rounds 4 to	16)				
Coeffs.	Food	Othfd	Nonfd				
β	-0.236*	-0.210*	0.447*				
	(-4.74)	(-2.50)	(4.61)				
Dun	umy coeffi	cients for to	tal exp.		Total exp	o. coefficier	nts
Coeffs.	Food	Othfd	Nonfd	Coeffs.	Food	Othfd	Nonfd
			Regime 2 (r	ounds 17 to	27)		
$d\beta_i^2$	0.118**	0.209*	-0.328*	$\beta_i^2$	-0.118*	-0.002	0.119**
•••	(1.96)	(3.72)	(-2.27)		(-2.29)	(-0.02)	(1.93)
			Regime 3 (r	ounds 42 to	47)		
$d\beta_i^3$	0.031*	-0.004**	-0.026	$\beta_i^3$	-0.206	-0.214	0.420**
	(2.43)	(-1.93)	(-0.10)		(-1.65)	(-1.03)	(1.76)

A.

63

Table B.3. Coefficients for Dynamic LA-AIDS Model: Urban (Contd.)

# CENTRE FOR DEVELOPMENT ECONOMICS WORKING PAPER SERIES

<u>No</u> .	Author(s)	Title
1	Kaushik Basu Arghya Ghosh Tridip Ray	The <u>Babu</u> and The <u>Boxwallah</u> : Managerial Incentives and Government Intervention (January 1994). <u>Review of Development Economics, 1997</u>
2	M.N. Murty Ranjan Ray	Optimal Taxation and Resource Transfers in a Federal Nation (February 1994)
3	V. Bhaskar Mushtaq Khan	Privatization and Employment : A Study of The Jute Industry in Bangladesh (March 1994). <u>American</u> Economic Review, March 1995, pp. 267-273
4	V. Bhaskar	Distributive Justice and The Control of Global Warming (March 1994) The North, the South and the Environment: V. Bhaskar and Andrew Glyn (Ed.) Earthscan Publication London, February 1995
5	Bishnupriya Gupta	The Great Depression and Brazil's Capital Goods Sector: A Re-examination (April 1994). <u>Revista Brasileria de</u> Economia 1997
6	Kaushik Basu	Where There Is No Economist: Some Institutional and Legal Prerequisites of Economic Reform in India (May 1994)
7	Partha Sen	An Example of Welfare Reducing Tariff Under Monopolistic Competition (May 1994), <u>Reveiw of</u> <u>International Economics, (forthcoming)</u>
8	Partha Sen	Environmental Policies and North-South Trade : A Selected Survey of the Issues (May 1994)
9	Partha Sen Arghya Ghosh Abheek Barman	The Possibility of Welfare Gains with Capital Inflows in A Small Tariff-Ridden Economy (June 1994)
10	V. Bhaskar	Sustaining Inter-Generational Altruism when Social Memory is Bounded (June 1994)
11	V. Bhaskar	Repeated Games with Almost Perfect Monitoring by Privately Observed Signals (June 1994)

No.	Author(s)	Title	No.
12	S. Nandeibam	Coalitional Power Structure in Stochastic Social Choice Functions with An Unrestricted Preference Domain (June 1994). Journal of Economic Theory (Vol. 68 No.	25 26
13	Kaushik Basu	1, January 1996, pp. 212-233 The Axiomatic Structure of Knowledge And Perception (July 1994)	27
14	Kaushik Basu	Bargaining with Set-Valued Disagreement (July 1994). Social Choice and Welfare, 1996, (Vol. 13, pp. 61-74)	<b>28</b> <sup>‡</sup>
15	S. Nandeibam	A Note on Randomized Social Choice and Random Dictatorships (July 1994). <u>Journal of Economic</u> Theory, Vol. 66, No. 2, August 1995, pp. 581-589	29
16	Mrinal Datta Chaudhuri	Labour Markets As Social Institutions in India (July 1994)	30
17	S. Nandeibam	Moral Hazard in a Principal-Agent(s) Team (July 1994) <u>Economic Design Vol. 1, 1995, pp. 227-250</u>	31
18	D. Jayaraj S. Subramanian	Caste Discrimination in the Distribution of Consumption Expenditure in India: Theory and Evidence (August 1994)	32
19	K. Ghosh Dastidar	Debt Financing with Limited Liability and Quantity Competition (August 1994)	2.
20	Kaushik Basu	Industrial Organization Theory and Developing Economies (August 1994). Indian Industry: Policies and Performance, D. Mookherjee (ed.), <u>Oxford</u> <u>University Press, 1995</u>	3:
21	Partha Sen	Immiserizing Growth in a Model of Trade with Monopolisitic Competition (August 1994). <u>The</u> <u>Review of International Economics, (forthcoming)</u>	9
22	K. Ghosh Dastidar	Comparing Cournot and Bertrand in a Homogeneous Product Market (September 1994)	d <sup>e</sup> ra
23	K. Sundaram S.D. Tendulkar	On Measuring Shelter Deprivation in India (September 1994)	
24	Sunil Kanwar	Are Production Risk and Labour Market Risk Covariant? (October 1994)	ţ

\*\*

Ĺ

No.	Author(s)	Title
25	Partha Sen	Welfare-Improving Debt Policy Under Monopolistic Competition (November 1994)
26	Ranjan Ray	The Reform and Design of Commodity Taxes in the presence of Tax Evasion with Illustrative Evidence from India (December 1994)
27	Wietze Lise	Preservation of the Commons by Pooling Resources, Modelled as a Repeated Game (January 1995)
28	Jean Drèze Anne-C. Guio Mamta Murthi	Demographic Outcomes, Economic Development and Women's Agency (May 1995). <u>Population and</u> Development Review, December, 1995
29	Jean Drèze Jackie Loh	Literacy in India and China (May 1995). Economic and Political Weekly, 1995
30	Partha Sen	Fiscal Policy in a Dynamic Open-Economy New- Keynesian Model (June 1995)
31	S.J. Turnovsky Partha Sen	Investment in a Two-Sector Dependent Economy (June 1995). <u>The Journal of Japanese and International Economics, Vol. 9, No. 1, March 1995</u>
32	K. Krishnamurty V. Pandit	India's Trade Flows: Alternative Policy Scenarios: 1995-2000 (June 1995). Indian Economic Review, Vol. 31, No. 1, 1996
33	Jean Drèze P.V. Srinivasan	Widowhood and Poverty in Rural India: Some Inferences from Household Survey Data (July 1995). Journal of Development Economics, 1997
34	Ajit Mishra	Hierarchies, Incentives and Collusion in a Model of Enforcement (January 1996)
35	Sunil Kanwar	Does the Dog wag the Tail or the Tail the Dog? Cointegration of Indian Agriculture with Non- Agriculture (February 1996)
36	Jean Drèze P.V. Srinivasan	Poverty in India: Regional Estimates, 1987-8 (February 1996)
37	Sunit Kanwar	The Demand for Labour in Risky Agriculture (April 1996)
38	Partha Sen	Dynamic Efficiency in a Two-Sector Overlapping Generations Model (May 1996)

5.2

< 7

\*\*\*\*<sup>2</sup>

No.	Author(s)	Title
39	Partha Sen	Asset Bubbles in a Monopolistic Competitive Macro Model (June 1996)
40	Pami Dua Stephen M. Miller David J. Smyth	Using Leading Indicators to Forecast US Home Sales in a Bayesian VAR Framework (October 1996)
41	Pami Dua David J. Smyth	The Determinants of Consumers' Perceptions of Buying Conditions for Houses (November 1996)
42	Aditya Bhattacharjea	Optimal Taxation of a Foreign Monopolist with Unknown Costs (January 1997)
43	M. Datta-Chaudhuri	Legacies of the Independence Movement to the Political Economy of Independent India (April 1997)
44	Suresh D. Tendulkar T. A. Bhavanì	Policy on Modern Small Scale Industries: A Case of Government Failure (May 1997)
45	Partha Sen	Terms of Trade and Welfare for a Developing Economy with an Imperfectly Competitive Sector (May 1997)
46	Partha Sen	Tariffs and Welfare in an Imperfectly Competitive Overlapping Generations Model (June 1997)
47	Pami Dua Roy Batchelor	Consumer Confidence and the Probability of Recession: A Markov Switching Model (July 1997)
48	V. Pandit B. Mukherji	Prices, Profits and Resource Mobilisation in a Capacity Constrained Mixed Economy (August 1997)
49	Ashwini Deshpande	Loan Pushing and Triadic Relations (September 1997)
50	Rinki Sarkar	Depicting the Cost Structure of an Urban Bus Transit Firm (September 1997)
51	Sudhir A. Shah	Existence and Optimality of Mediation Schemes for Games with Communication (November 1997)
52	V. Pandit	A Note on Data Relating to Prices in India (November 1997)
53	Rinki Sarkar	Cost Function Analysis for Transportation Modes: A Survey of Selective Literature (December 1997)

•

. 29

\$}

5.25r

No.	Author(s)	Title
54	Rinki Sarkar	Economic Characteristics of the Urban Bus Transit Industry: A Comparative Analysis of Three Regulated Metropolitan Bus Corporations in India (February 1998)
55	Aditya Bhattacharjea	Was Alexander Hamilton Right? Limit-pricing Foreign Monopoly and Infant-industry Protection (February 1998)
56	Bishwanath Goldar Badal Mukherji	Pollution Abatement Cost Function: Methodological and Estimation Issues (March 1998)
57	Smita Misra	Economies of Scale in Water Pollution Abatement: A Case of Small-Scale Factories in an Industrial Estate in India (April 1998)
58	T.A. Bhavani Suresh D. Tendulkar	Determinants of Firm-level Export Performance: A Case Study of Indian Textile Garments and Apparel Industry (May 1998)
59	Partha Sen	Non-Uniqueness In The First Generation Balance of Payments Crisis Models (December 1998)
60	Ranjan Ray J.V. Meenakshi	State-Level Food Demand in India: Some Evidence on Rank-Three Demand Systems (December 1998)
61	Brinda Viswanathan	Structural Breaks in Consumption Patterns: India, 1952- 1991 (December 1998)

ć.

Â.

69

y 18